Author’s response to reviews

Title: A novel 11beta-hydroxysteroid dehydrogenase type1 inhibitor CNX-010-49 improves hyperglycemia, lipid profile and reduces body weight in diet induced obese C57B6/J mice with a potential to provide cardio protective benefits

Authors:

Tharappel M Anil (anil.t.m@connexios.com)
Anilkumar Dandu (anildandu@gmail.com)
Krishnareddy Harsha (harsha.k@connexios.com)
Jaideep Singh (jaideepsingh21@gmail.com)
Nitya Shree (nityashree@connexios.com)
Venkatesh SatishKumar (satishkumar.v@connexios.com)
Mudigere N Lakshmi (lakshmi.m@connexios.com)
Venkategowda Sunil (ysunilgowda28@gmail.com)
Chandrashekaran Harish (harish.c@connexios.com)
Gundalmandikal V Balamurali (muraligv.la@gmail.com)
Baisani S NaveenKumar (navenkumar@connexios.com)
Aralakuppe S Gopala (gopala.a.s@connexios.com)
Shivakumar Pratibha (pratibha.s@connexios.com)
ManojKumar Sadasivuni (manojsadas@gmail.com)
Mammen O Anup (anup.oommen@connexios.com)
Y Moolemath (moolemath@connexios.com)
Marikunte V Venkataraman (m.v.venkataraman@connexios.com)
Madanahalli R Jagannath (m.r.jagannath@connexios.com)
Baggavalli P Somesh (somesh.baggavalli@connexios.com)

Version: 2
Date: 7 February 2014

Author’s response to reviews:

Acknowledgement section has been added as suggested